Author’s response to reviews

Title: Primary vaginal calculus in a woman with urogenital sinus anomaly: A case report

Authors:

Qiong Xu (5512040@zju.edu.cn)

yu zou (zouyuzju@zju.edu.cn)

Version: 1 Date: 29 Apr 2020

Author’s response to reviews:

Dear, Anton Brostedt,

Thank you for those comments.
I have revised the discussion, and carry out the corrections

Best wishes,

Yu zou